Author’s response to reviews

Title: Women’s empowerment and fertility preferences in high fertility countries in Sub-Saharan Africa

Authors:

Esso - Hanam ATAKE (atakesyl@gmail.com)
Pitaloumani GNAKOU ALI (gnak2005@yahoo.fr)

Version: 4 Date: 11 Mar 2019

Author’s response to reviews:

We are very grateful to the editors for their appropriate and constructive suggestions and for their proposed corrections to improve the paper. We have addressed all the issues raised and have modified the paper accordingly.

i) Table 2: The authors should cross-check the distributions by husband’s education level for Niger and Chad, as well as the distributions by religion for Burkina Faso and Chad as the numbers do not seem to add up.

We revised the distribution by husband’s education level for Niger and Chad, as well as the distributions by religion for Burkina Faso and Chad (Please, refer to Table 2).

(ii) All spelling and grammatical errors as well as suggested rephrasing pointed out have been corrected. Thank you very much.